Author's response to reviews

Title: A Randomized Trial of an Intervention to Improve Use and Adherence to Effective Heart Disease Prevention Strategies

Authors:

Stacey L Sheridan (sls593@med.unc.edu)
Lindy B Draeger (lindy@unc.edu)
Michael P Pignone (pignone@med.unc.edu)
Thomas C Keyserling (jato@med.unc.edu)
Ross J Simpson Jr (rsimpson@med.unc.edu)
Barbara Rimer (brimer@email.unc.edu)
Shrikant I Bangdiwala (kant@unc.edu)
Jianwen Cai (cai@bios.unc.edu)
Ziya Gizlice (Ziya_gizlice@unc.edu)

Version: 2 Date: 28 October 2011

Author's response to reviews: see over
Editors
BMC Health Services Research

Dear Editors,

We are pleased to re-submit our manuscript “A Randomized Trial of an Intervention to Improve Use and Adherence to Effective Heart Disease Prevention Strategies” for further consideration for publication. We appreciate the thoughtful comments by you and our reviewers and think they have helped to strengthen the manuscript. These comments and our responses are in the attached reviewer response grid.

We appreciate your additional consideration of our work and ask that you address me, Stacey Sheridan, with any further correspondence.

Thank you,

Stacey L. Sheridan, MD, MPH
Associate Professor of Medicine